Author's response to reviews

Title: Moraxella catarrhalis acquisition, airway inflammation and protease-antiprotease balance in chronic obstructive pulmonary disease

Authors:

Ganapathi I Parameswaran (gp36@buffalo.edu)
Catherine T Wrona (wrona@buffalo.edu)
Timothy F Murphy (murphyt@buffalo.edu)
Sanjay Sethi (ssethi@buffalo.edu)

Version: 4 Date: 6 November 2009

Author's response to reviews:

Cover letter for resubmitted manuscript “Moraxella catarrhalis acquisition, airway inflammation and protease-antiprotease balance in chronic obstructive pulmonary disease”.

To Nov 6, 2009
The Editorial team
BMC Infectious Diseases

Thank you for accepting our manuscript for publication. We have made the following changes as suggested.
1. Author names written as in the submission system.
2. Author affiliations re-written with addresses.
3. Author contributions include the sentence “All authors read and approved the final manuscript”.
4. The order of Figures has been changed so that they are consecutively mentioned in the manuscript.
5. Author qualifications have been removed.
6. Underlining removed.
7. Introduction changed to Background, and Methods and Materials to Methods.
8. Abbreviation formatted as a paragraph.
9. Competing interests section completed.
10. Vertical lines in Tables have been removed.
11. Blank page has been removed.
12. Image files changed- titles removed, white space minimized.

Ganapathi I Parameswaran
Catherine T Wrona
Timothy F Murphy
Sanjay Sethi